*

0 2 4 6 8 (a) (b) (c)

0,0 0,5 1,0 2

4

6

8 (d) lo g(1 0^k

P^ω )

0,0 0,5 1,0

(e)

0,0 0,5 1,0 ω/ω_0

(f)

Fig.5. Power spectra associated with transient frequency locking for L=8 and L=21. Power spectra are shown at the following times: (a) and (d) from t=100 to t=646.13; (b) and (e) from t=1500 to t=2046.13; (c) and (f) from t=20000 to t=20546.13, respectively.

0,0 0,5 1,0 0 2 4 6 8 10 (a)

log (10 kP^ω )

0,0 0,5 1,0

(b)

0,0 0,5 1,0 1,5 ω/ω_0

(c)

Fig.6 Power spectra in the region of the lowest window of period 5 and its neighborhood for ω0=2p. (a) L=30.5 (quasiperiodic solution); (b) L=31.8 (frequency locking of period 5); (c) L=33.0 (quasiperiodic solution). Power spectra were calculated in the time interval from t=20000 to t=20546.13.